% version 2013
\documentclass[11pt]{amsart}
%\usepackage{graphicx, color}
\usepackage{amscd}
\usepackage{amsmath}
\usepackage{amsfonts}
\usepackage{amssymb}
\usepackage{mathrsfs}

\textwidth=6in \textheight=9.5in \topmargin=-0.5cm
\oddsidemargin=0.5cm \evensidemargin=0.5cm
%\usepackage[notref,notcite]{showkeys}

\newtheorem{theorem}{Theorem}
\newtheorem{ex}{Example}
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{prop}[theorem]{Proposition}
\newtheorem{remark}{Remark}
\newtheorem{corollary}[theorem]{Corollary}
\newtheorem{claim}{Claim}
\newtheorem{step}{Step}
\newtheorem{case}{Case}
\newtheorem{definition}[theorem]{Definition}

%\newenvironment{proof}[1][Proof]{\textbf{#1.} }{\ \rule{0.5em}{0.5em}}
\newenvironment{proof-sketch}{\noindent{\bf Sketch of Proof}\hspace*{1em}}{\qed\bigskip}



%\numberwithin{equation}{section}
\newcommand{\RR}{\mathbb R}
\newcommand{\NN}{\mathbb N}
\newcommand{\PP}{\mathbb P}
\newcommand{\ZZ}{\mathbb Z}
\renewcommand{\le}{\leqslant}
\renewcommand{\leq}{\leqslant}
\renewcommand{\ge}{\geqslant}
\renewcommand{\geq}{\geqslant}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\baselineskip=16pt plus 1pt minus 1pt

\begin{document}
\begin{thebibliography}{99}

\bibitem{1} {\sc J.I. Diaz, J.E. Saa},  Existence et unicit\'{e} de solutions positives pour certaines \'equations elliptiques quasilin\'eaires, {\it C.R. Acad. Sci. Paris} {\bf 305} (1987), 521-524.
\bibitem{2} {\sc L. Gasinski, N.S. Papageorgiou}, {\em Nonlinear Analysis}, Chapman $\&$ Hall/CRC, Boca Raton, 2006.
\bibitem{3} {\sc D. Mugnai, N.S. Papageorgiou}, Resonant nonlinear Neumann problems with indefinite weight, {\it  Annali Sc. Norm. Sup. Pisa Cl. Sci.} {\bf 11} (2012), 729-788.
\bibitem{4} {\sc N.S. Papageorgiou, V.D. R\u{a}dulescu}, Semilinear Neumann problems with indefinite and unbounded potential and crossing nonlinearity, {\it Contemp. Math.} {\bf 595} (2013), 293-316.
\bibitem{5} {\sc N.S. Papageorgiou, G. Smyrlis},  On a class of parametric Neumann problems with indefinite and unbounded potential, {\it Forum Math.}, DOI:10.1515/forum-2012-0042.
\bibitem{6} {\sc J.L. Vazquez},  A strong maximum principle for some quasilinear elliptic equations, {\it Appl. Math. Optim.} {\bf 12} (1984), 191-202.
\bibitem{7} {\sc X. Wang}  Neumann problems of semilinear elliptic equations involving critical Sobolev exponents, {\it J. Differential Equations} {\bf 93} (1991), 283-310.




\end{thebibliography}

%\hfill\today\bigskip

\title[Positive solutions for Neumann problems]{Positive solutions for Neumann problems with indefinite and unbounded potential}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\author[N.S. Papageorgiou]{Nikolaos S. Papageorgiou}
\address{National Technical University, Department of Mathematics,
				Zografou Campus, Athens 15780, Greece}
\email{\tt npapg@math.ntua.gr}

\author[V. R\u{a}dulescu]{Vicen\c{t}iu D. R\u{a}dulescu}
\address{Institute of Mathematics "Simion Stoilow" of the Romanian Academy, P.O. Box 1-764,
          014700 Bucharest, Romania \\
          and Department of Mathematics, University of Craiova, Street A.I.Cuza 13,
          200585 Craiova, Romania}
\email{\tt vicentiu.radulescu@imar.ro}

\keywords{Positive solutions, maximum principle, Picone's identity, indefinite and unbounded potential, logistic equation.\\
\phantom{aa} 2010 AMS Subject Classification:
35J20, 35J60}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


\begin{abstract}
We consider semilinear Neumann equations with an indefinite and unbounded potential. We establish the existence and uniqueness of positive solutions. We show that our setting incorporates as special cases several parametric equations of interest (such as the equidiffusive logistic equation).
\end{abstract}

\maketitle

%\tableofcontents

\section{Introduction}

In this paper we deal with the following semilinear Neumann problem
\begin{eqnarray}\label{eq1}
	-\Delta u(z)+\beta(z)u(z)=f(z,u(z))\ \mbox{in}\ \Omega,\quad \frac{\partial u}{\partial n}=0\ \mbox{on}\ \partial\Omega.
\end{eqnarray}

Here $\Omega\subseteq\RR^N$ is a bounded domain with a $C^2$-boundary, $\beta$ is a potential function which is in general unbounded and sign changing. The reaction $f(z,x)$ is a Carath\'eodory function (that is, for all $x\in\RR$ the mapping $z\longmapsto f(z,x)$ is measurable and for a.a. $z\in\Omega$ $x\longmapsto f(z,x)$ is continuous), which exhibits general growth conditions near $+\infty$ and near $0^+$. As we will see these conditions incorporate in our framework as special cases various paramettric problems such as equidiffusive logistic equations. We are interested in the existence and uniqueness of positive solutions.

Recently semilinear Neumann problems with unbounded and indefinite potential, were studied by Papageorgiou and R\u{a}dulescu \cite{4}. They deal with equations in which the reaction $f(z,x)$ exhibits an asymmetric behavior at $+\infty$ and at $-\infty$ (jumping nonlinearrity) and they prove multiplicity theorems providing sign information for all the solutions. We mention also the recent works of Mugnai and Papageorgiou \cite{3}, who examine equations driven by the $p$-Laplacian plus an indefinite potential and of Papageorgiou and Smyrlis \cite{5}, who consider a special class of coercive semilinear equations.

\section{Positive Solutions}

In the analysis of problem (\ref{eq1}) we will use the Sobolev space $H^1(\Omega)$ and the ordered Banach space $C^1(\overline{\Omega})$. The positive cone of the latter space is given by
$$C_+=\{u\in C^1(\overline{\Omega}): u(z)\geq 0\ \mbox{for all}\ z\in \overline{\Omega}\}.$$

This cone has a nonempty interior
$$\mbox{int}\, C_+=\{u\in C_+:u(z)>0\ \mbox{for all}\ z\in \overline{\Omega}\}.$$

The hypotheses on the potential function $\beta(\cdot)$ are the following:

\smallskip
$H(\beta):$ $\beta\in L^s(\Omega)$ with $s>\frac{N}{2}$ and $\beta^+\in L^{\infty}(\Omega)$

\smallskip
By $\{\hat{\lambda}_n\}_{n\geq 1}$ we denote the distinct eigenvalues of the differential operator $u\rightarrow-\Delta u+\beta(z)u$, $u\in H^1(\Omega)$. We know that $\hat{\lambda}_1$ is simple and
\begin{eqnarray}\label{eq2}
	\hat{\lambda}_1=\inf\left[\frac{\xi(u)}{||u||^{2}_{2}}:u\in H^1(\Omega),\ u\neq 0\right]
\end{eqnarray}
where $\xi(u)=||Du||^{2}_{2}+\int_{\Omega}\beta(z)u^2dz$ for all $u\in H^1(\Omega)$. The infimum in (\ref{eq2}) is realized on the one dimensional eigenspace corresponding to $\hat{\lambda}_1$. From (\ref{eq2}) it is clear that the elements of this eigenspace do not change sign. By $\hat{u}_1$ we denote the positive $L^2$-normalized (that is, $||\hat{u}_1||_2=1$) eigenfunction corresponding to $\hat{\lambda}_1$. The regularity results of Wang \cite{7} imply that $\hat{u}_1\in C_+\backslash\{0\}$ and in fact using $H(\beta)$ and the maximum principle of Vazquez \cite{6}, we have $\hat{u}_1\in \mbox{int}\, C_+$. The next lemma is an consequence of these properties (see Papageorgiou and R\u{a}dulescu \cite{4}).
\begin{lemma}\label{lem1}
	If $\vartheta\in L^{\infty}(\Omega)$ and $\vartheta(z)\leq\hat{\lambda}_1$ a.e. in $\Omega$, $\vartheta\neq\hat{\lambda}_1$, then there exists $c_0>0$ such that
	$$\xi(u)-\int_{\Omega}\vartheta(z)u^2dz\geq c_0||u||^2\ \mbox{for all}\ u\in H^1(\Omega).$$
\end{lemma}

The hypotheses on the reaction $f(z,x)$ are the following:

\smallskip
\textbf{$H(f):$} $f:\Omega\times\RR\rightarrow\RR$ is a Carath\'eodory function such that  $f(z,0)=0$ for a.a. $z\in \Omega$ and
\begin{itemize}
	\item[(i)] $|f(z,x)|\leq a(z)(1+x^{r-1})$ for a.a. $z\in\Omega$, all $x\geq 0$ with $a\in L^{\infty}(\Omega)_+$, $2<r<2^*=\left\{
	\begin{array}{cl}
		\frac{2N}{N-2}&\mbox{if}\ 3\leq N\\
		+\infty&\mbox{if}\ N=1,2
	\end{array}
	\right.;$
	\item[(ii)] $\limsup\limits_{x\rightarrow+\infty}\frac{f(z,x)}{x}\leq\vartheta(z)$ uniformly for a.a. $z\in\Omega$, with $\vartheta\in L^{\infty}(\Omega)$, $\vartheta(z)\leq\hat{\lambda}_1$ a.e in $\Omega$, $\vartheta\not\equiv\hat{\lambda}_1$;
	\item[(iii)] $\liminf\limits_{x\rightarrow 0^+}\frac{f(z,x)}{x}\geq\eta(z)$ uniformly for a.a. $z\in\Omega$, with $\eta\in L^{\infty}(\Omega)$, $\eta(z)\geq\hat{\lambda}_1$ a.e. in $\Omega$, $\eta\not\equiv\hat{\lambda}_1$.
\end{itemize}
\begin{remark}
	Since we are interested in positive solutions and the above hypotheses concern the positive semi-axis $\RR_+=\left[0,+\infty\right)$, without any loss of generality we assume that $f(z,x)=0$ for a.a. $z\in\Omega$, all $x\leq 0$. Note that according to hypothesis $H(f)(i)$, $f(z,\cdot)$ has subcritical growth. Finally hypotheses $H(f)(ii),(iii)$ imply that the quotient $\frac{f(z,x)}{x}$ crosses at least the principal eigenvalue $\hat{\lambda}_1$ as we move from $0^+$ to $+\infty$.
\end{remark}

>From the spectral analysis of Papageorgiou and R\u{a}dulescu \cite{4} we know that there exists $\gamma_0>\max\{-\hat{\lambda}_1,1\}$ such that
\begin{eqnarray}\label{eq3}
	\xi(u)+\gamma_0||u||^{2}_{2}\geq c_1||u||^2\ \mbox{for all}\ u\in H^1(\Omega),\ \mbox{some}\ c_1>0.
\end{eqnarray}

We introduce the Carath\'eodory function $\hat{f}(z,x)=\left\{
\begin{array}{cl}
	0&\mbox{if}\ x\leq 0\\
	f(z,x)+\gamma_0x&\mbox{if}\ 0<x
\end{array}
\right.$ and its primitive $\hat{F}(z,x)=\int^{x}_{0}\hat{f}(z,s)ds$.
\begin{prop}\label{prop2}
	If hypotheses $H(\beta)$ and $H(f)$ hold, then problem (\ref{eq1}) has at least one positive solution $u_0\in \mbox{int}\, C_+$.
\end{prop}
\begin{proof}
	Let $\hat{\varphi}:H^1(\Omega)\rightarrow\RR$ be the $C^1$-functional defined by $$\hat{\varphi}(u)=\frac{1}{2}\xi(u)+\frac{\gamma_0}{2}||u||^{2}_{2}-\int_{\Omega}\hat{F}(z,u)dz\ \mbox{for all}\ u\in H^1(\Omega).$$
	
	Hypotheses $H(f)(i),(ii)$ imply that given $\epsilon>0$, we can find $c_2=c_2(\epsilon)>0$ such that
	\begin{eqnarray}\label{eq4}
		F(z,x)\leq\frac{1}{2}(\vartheta(z)+\epsilon)(x^+)^2+c_2\ \mbox{for a.a.}\ z\in\Omega,\ \mbox{all}\ x\in\RR
	\end{eqnarray}
	(recall that for all $x\in\RR$, $x^{\pm}=\max\{\pm x,0\}$). Then
	\begin{eqnarray*}
		 &&\hat{\varphi}(u)\geq\frac{1}{2}\left[\xi(u^+)-\int_{\Omega}\vartheta(z)(u^+)^2dz\right]-\frac{\epsilon}{2}||u^+||^2+\frac{1}{2}\xi(u^-)+\\
		&&\hspace{3.3cm}+\frac{\gamma_0}{2}||u^-||^{2}_{2}-c_2|\Omega|_N\ (\mbox{see (\ref{eq4})}),
	\end{eqnarray*}
	where $u^{\pm}(\cdot)=u(\cdot)^{\pm}$ and $|\cdot|_N$ denotes the Lebesgue measure on $\RR^N$. Using Lemma \ref{lem1} and (\ref{eq3}), we obtain
	 $$\hat{\varphi}(u)\geq\frac{c_0-\epsilon}{2}||u^{+}||^2+\frac{c_1}{2}||u^-||^2-c_2|\Omega|_N.$$
	
	Choosing $\epsilon\in(0,c_0)$, we see that $\hat{\varphi}$ is coercive. Also, using the Sobolev embedding theorem, we check that $\hat{\varphi}$ is sequentially weakly lower semicontinuous. So, we can find $u_0\in H^1(\Omega)$ such that
	\begin{eqnarray}\label{eq5}
		\hat{\varphi}(u_0)=\inf[\hat{\varphi}(u):u\in H^1(\Omega)]
	\end{eqnarray}
	
	Hypothesis $H(f)_1(iii)$ implies that given $\epsilon>0$, we can find $\delta=\delta(\epsilon)>0$ such that
	\begin{eqnarray}\label{eq6}
		F(z,x)\geq\frac{1}{2}(\eta(z)-\epsilon)x^2\ \mbox{for a.a.}\ z\in\Omega,\ \mbox{all}\ x\in[0,\delta]
	\end{eqnarray}
	
	Let $t\in(0,1)$ be small such that  $t\hat{u}_1(z)\in\left(0,\delta\right]$ for all $z\in\overline{\Omega}$ (recall that $\hat{u}_1\in \mbox{int}\, C_+$). Then
	\begin{eqnarray*}
		 &&\hat{\varphi}(t\hat{u}_1)=\frac{t^2}{2}\xi(\hat{u}_1)-\int_{\Omega}F(z,t\hat{u}_1)dz\leq\frac{t^2}{2}\int_{\Omega}(\hat{\lambda}_1-\eta(z))\hat{u}_1^2dz+\frac{t^2\epsilon}{2}\\
		&&\hspace{3cm}(\mbox{see (\ref{eq6}) and recall that $||\hat{u}_1||_2=1$})
	\end{eqnarray*}
	
	Note that $\hat{\mu}=\int_{\Omega}(\eta(z)-\hat{\lambda}_1)\hat{u}_1^2dz>0$ (see hypothesis $H(f)(iii)$). Hence
	$$\hat{\varphi}(t\hat{u}_1)\leq\frac{t^2}{2}[-\hat{\mu}+\epsilon]$$
	
	So, choosing $\epsilon\in(0,\hat{\mu})$, we see that $\hat{\varphi}(t\hat{u}_1)<0$. Therefore
	$$\hat{\varphi}(u_0)<0=\hat{\varphi}(0)\ \mbox{(see (\ref{eq5})), hence}\ u_0\neq 0.$$
	
	From (\ref{eq5}) we have $\hat{\varphi}'(u_0)=0$ and so
	\begin{eqnarray}\label{eq7} &&\int_{\Omega}(Du_0,Dh)_{\RR^N}dz+\int_{\Omega}\beta(z)u_0hdz+\gamma_0\int_{\Omega}u_0hdz=\int_{\Omega}\hat{f}(z,u_0)hdz\ \mbox{for all}\ h\in H^1(\Omega).
	\end{eqnarray}
	
	In (\ref{eq7}) we choose $h=-u^{-}_{0}\in H^1(\Omega)$. Then
	$$\xi(u^{-}_{0})+\gamma_0||u^{-}_{0}||^{2}_{2}=0,\ \mbox{hence}\ u_0\geq 0,\ u_0\neq 0\ (\mbox{see (\ref{eq3})})$$
	
	Then from (\ref{eq7}) and Green's identity we obtain
	$$-\Delta u_0(z)+\beta(z)u_0(z)=f(z,u_0(z))\ \mbox{a.e. in}\ \Omega,\ \frac{\partial u_0}{\partial n}=0\ \mbox{on}\ \partial\Omega.$$
	
	From Wang \cite{7} we obtain $u_0\in C_+\backslash\{0\}$. Note that hypotheses $H(f)(i),(iii)$ imply that given $\rho>0$, we can find $\xi_{\rho}>0$ such that  $f(z,x)+\xi_{\rho}x \geq 0$ for a.a. $z\in \Omega$, all $x\in[0,\rho]$. If $\rho=||u_0||_{\infty}$ and $\xi_{\rho}>0$ as above, then
	\begin{eqnarray*}
		&&-\Delta u_0(z)+(\beta(z)+\xi_{\rho})u_0(z)\geq 0\ \mbox{a.e. in}\ \Omega,\\
		&\Rightarrow&\Delta u_0(z)\leq(||\beta^+||_{\infty}+\xi_{\rho})u_0(z)\ \mbox{a.e. in}\ \Omega,\\
		&\Rightarrow&u_0\in \mbox{int}\, C_+\ (\mbox{see Vazquez \cite{6}}).
	\end{eqnarray*}
\end{proof}

If we strengthen the conditions on $f(z,\cdot)$ we can assure uniqueness of the positive solution. The new stronger hypotheses on the reaction $f(z,x)$ are the following:

\smallskip
\textbf{$H(f)':$} $f:\Omega\times\RR\rightarrow\RR$ is a Carath\'eodory function such that  $f(z,0)=0$ for a.a. $z\in\Omega$, hypotheses $H(f)'$\\
$(i),(ii),(iii)$ are the same as the corresponding hypotheses $H(f)(i),(ii),(iii)$ and\\ $(iv)$ for a.a. $z\in \Omega$ $x\rightarrow\frac{f(z,x)}{x}$ is decreasing and for all $z\in\Omega_0\subseteq\Omega$ with $|\Omega_0|_N>0$ $x\rightarrow\frac{f(z,x)}{x}$ is strictly decreasing on $(0,+\infty)$.
\begin{prop}\label{prop3}
	If hypotheses $H(\beta)$ and $H(f)'$ hold, then problem (\ref{eq1}) admits a unique positive solution $u_0\in \mbox{int}\, C_+$.
\end{prop}
\begin{proof}
	From Proposition \ref{prop2}, we already have one solution $u_0\in \mbox{int}\, C_+$. Let $y$ be another positive solution of (\ref{eq1}). As before we can show that $y\in \mbox{int}\, C_+$.
	
	We consider the integral functional $\tau:L^2(\Omega)\rightarrow\overline{\RR}=\RR\cup\{+\infty\}$ defined by
	
	$$\tau(u)=\left\{
	\begin{array}{cl}
		\frac{1}{2}||Du^{1/2}||^{2}_{2}+\frac{1}{2}\int_{\Omega}\beta(z)udz&\mbox{if}\ u\geq 0,\ u^{1/2}\in H^1(\Omega)\\
		+\infty&\mbox{otherwise}
	\end{array}
	\right.$$
	
	From Lemma 1 of Diaz and Saa \cite{1}, we know that $\tau(\cdot)$ is convex and lower semicontinuous. Since $u_0,y\in \mbox{int}\, C_+$, we see that given any $h\in C^1(\overline{\Omega})$ for $|\lambda|\leq 1$ small we have
	\begin{eqnarray*}
		&&u_0^2+\lambda h,\ y^2+\lambda h\in dom\tau=\{u\in H^1(\Omega):\tau(u)<+\infty\}\\
		&&\hspace{5cm}(\mbox{effective domain of}\ \tau)
	\end{eqnarray*}
	
	Therefore $\tau$ is G\^ateaux differentiable at $u_0^2$ and at $y^2$ in the direction $h$. Moreover, using the chain rule, we have
	\begin{eqnarray*}
		&&\tau'(u_0^2)(h)=\frac{1}{2}\int_{\Omega}\frac{-\Delta u_0}{u_0}hdz+\frac{1}{2}\int_{\Omega}\beta(z)hdz\\
		&&\tau'(y^2)(h)=\frac{1}{2}\int_{\Omega}\frac{-\Delta y}{y}hdz+\frac{1}{2}\int_{\Omega}\beta(z)hdz\ \mbox{for all}\ h\in H^1(\Omega)
	\end{eqnarray*}
	(recall that $C^1(\overline{\Omega})$ is dense in $H^1(\Omega)$). The convexity of $\tau$ implies the monotonicity of $\tau'$. Therefore
	\begin{eqnarray*}
		 &&0\leq\frac{1}{2}\int_{\Omega}\left(\frac{f(z,u_0)}{u_0}-\frac{f(z,y)}{y}\right)(u_0^2-y^2)dz\leq 0,\\
		&\Rightarrow&u_0=y\ (\mbox{see hypothesis}\ H(f)'(iv)\ \mbox{and recall that}\ u_0,y\in \mbox{int}\, C_+)
	\end{eqnarray*}
\end{proof}


\section{Special Cases}

First consider the following logistic equation with equidiffusive reaction
\begin{eqnarray}\label{eq8}
	-\Delta u(z)+\beta(z)u(z)=\lambda u(z)-h(z,u(z))\ \mbox{in}\ \Omega,\ \frac{\partial u}{\partial n}=0\ \mbox{on}\ \partial\Omega
\end{eqnarray}

The hypotheses on the perturbation $h(z,x)$ are the following:

\smallskip
\textbf{$H(h)_1:$} $h:\Omega\times\RR\rightarrow\RR$ is Carath\'eodory function such that  $h(z,0)=0$ for a.a. $z\in\Omega$ and
\begin{itemize}
	\item[(i)] $|h(z,x)|\leq a(z)(1+x^{r-1})$ for a.a. $z\in \Omega$, all $x\geq 0$, with $a\in L^{\infty}(\Omega)$ and $2<r<2^*$;
	\item[(ii)] $\lim\limits_{x\rightarrow+\infty}\frac{h(z,x)}{x}=+\infty$ uniformly for a.a. $z\in\Omega$;
	\item[(iii)] $\lim\limits_{x\rightarrow0^+}\frac{h(z,x)}{x}=0$ uniformly for a.a. $z\in\Omega$;
	\item[(iv)] for a.a. $z\in\Omega$ $x\rightarrow\frac{h(z,x)}{x}$ is increasing and for all $z\in\Omega_0\subseteq\Omega$ with $|\Omega_0|_N>0$ $x\rightarrow\frac{h(z,x)}{x}$ is strictly increasing.
\end{itemize}
\begin{remark}
	A typical perturbation $h(z,x)$ satisfying hypotheses $H(f)_1$ is $h(z,x)=h(x)=x^{r-1}$ with $2<r<2^*$, which corresponds to the classical equidiffusive logistic equation.
\end{remark}
\begin{prop}\label{prop4}
	If hypotheses $H(\beta)$ and $H(f)_1$ hold and $\lambda>\hat{\lambda}_1$, then problem (\ref{eq8}) admits a unique positive solution $u_0\in \mbox{int}\, C_+$.
\end{prop}

Now consider the following nonhomogeneous eigenvalue problem:
\begin{eqnarray}\label{eq9}
	-\Delta u(z)+\beta(z)u(z)=\lambda u(z)^{q-1}\ \mbox{in}\ \Omega,\ \frac{\partial u}{\partial n}=0\ \mbox{on}\ \partial\Omega.
\end{eqnarray}
\begin{prop}\label{prop5}
	If hypotheses $H(\beta)$ hold, $q\in(2,2^*)$ and $\lambda>0$, then problem (\ref{eq9}) has a unique positive solution $u_0\in \mbox{int}\, C_+$.
\end{prop}

Finally consider the following parametric Neumann problem:
\begin{eqnarray}\label{eq10}
	-\Delta u(z)+\beta(z)u(z)=\lambda u(z)+h(z,u(z))\ \mbox{in}\ \Omega,\ \frac{\partial u}{\partial n}=0\ \mbox{on}\ \partial\Omega.
\end{eqnarray}

The hypotheses on the perturbation $h(z,x)$ are the following:

\smallskip
\textbf{$H(h)_2:$} $h:\Omega\times\RR\rightarrow\RR$ is a Carath\'eodory function such that  for a.a. $z\in\Omega$ $h(z,0)=0$, $h(z,x)>0$ for all $x>0$ and
\begin{itemize}
	\item[(i)] $h(z,x)\leq a(z)(1+x^{p-1})$ for a.a. $z\in\Omega$, all $x\geq 0$ with $a\in L^{\infty}(\Omega)_+$;
	\item[(ii)] $\lim\limits_{x\rightarrow+\infty}\frac{h(z,x)}{x}=0$ uniformly for a.a. $z\in\Omega$;
	\item[(iii)] $\lim\limits_{x\rightarrow 0^+}\frac{h(z,x)}{x}=+\infty$ uniformly for a.a. $z\in\Omega$;
	\item[(iv)] for a.a. $z\in\Omega$, $x\rightarrow\frac{h(z,x)}{x}$ is decreasing and for all $z\in\Omega_0\subseteq\Omega$ with $|\Omega_0|_N>0$ $x\rightarrow\frac{h(z,x)}{x}$ is strictly decreasing on $(0,+\infty)$.
\end{itemize}
\begin{prop}\label{prop6}
	If hypotheses $H(\beta)$ and $H(h)_2$ hold and $\lambda<\hat{\lambda}_1$, then problem (\ref{eq10}) admits a unique positive solution $u_0\in \mbox{int}\, C_+$.
\end{prop}
In fact we can show that this upper bound $\hat{\lambda}_1$ is sharp.

\begin{prop}\label{prop7}
	If hypotheses $H(\beta)$ and $H(h)_2$ hold and $\lambda\geq\hat{\lambda}_1$, then problem (\ref{eq10}) has no positive solution.
\end{prop}
\begin{proof}
	Let $\lambda\geq\hat{\lambda}_1$ and suppose that problem (\ref{eq10}) has a positive solution $u_{\lambda}$. As in the proof of Proposition \ref{prop2}, we can show that $u_{\lambda}\in \mbox{int}\, C_+$. Let
	 $$R(\hat{u}_1,u_{\lambda})(z)=\left|D\hat{u}_1(z)\right|^2-\left|Du_{\lambda}(z)\right|^{p-2}\left(Du_{\lambda}(z),D\left(\frac{\hat{u}_1^2}{u_{\lambda}}\right)\right)_{\RR^N}$$
	
	From Picone's identity (see, for example, Gasinski and Papageorgiou \cite[p. 785]{2}, we have
	\begin{eqnarray*}
		 0\leq\int_{\Omega}R(\hat{u}_1,u_{\lambda})dz&=&||D\hat{u}_1||^{2}_{2}-\int_{\Omega}(-\Delta u_{\lambda})\frac{\hat{u}_1^2}{u_{\lambda}}dz\ (\mbox{by Green's identity})\\
		 &=&||D\hat{u}_1||^{2}_{2}-\int_{\Omega}(\lambda-\beta(z))\hat{u}_1^2dz-\int_{\Omega}h(z,u_{\lambda})\frac{\hat{u}_1^2}{u_{\lambda}}dz\\
		&<&(\hat{\lambda}_1-\lambda)\leq 0
	\end{eqnarray*}
	a contradiction.
\end{proof}
\begin{remark}
	Analogous results can be obtained for nonlinear equations driven by the $p$-Laplacian.
\end{remark}






\end{document} 